Author's response to reviews

Title: ADHD and Disruptive behavior scores - associations with MAO-A and 5-HTT genes and with platelet MAO-B activity in adolescents

Authors:

Kerstin Malmberg (kerstin.malmberg@ki.se)
Hanna-Linn Wargelius (hanna-linn.wargelius@neuro.uu.se)
Paul Lichtenstein (paul.lichtenstein@ki.se)
Lars Oreland (lars.oreland@neuro.uu.se)
Jan-Olov Larsson (jan-olov.larsson@ki.se)

Version: 8 Date: 11 April 2008

Author's response to reviews: see over
Re MS: 4344947721441078 - ADHD and Disruptive behavior scores - associations with MAO-A and 5-HTT genes and with platelet MAO-B activity in adolescents

Dear Editor,

On behalf of the co-authors and ourselves, we hereby resubmit our paper “ADHD and Disruptive behavior scores - associations with MAO-A and 5-HTT genes and with platelet MAO-B activity in adolescents” after formatting the changes.

We have now insert the Figure legend after the references and we hope the problem with endnote is solved, if not let us now.

Yours Sincerely,

Dr. Kerstin Malmberg and Hanna-Linn Wargelius